\documentclass[12pt]{umj_eng3}
\usepackage[T1,T2A]{fontenc}
\usepackage[english]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts, dsfont, graphicx, srcltx}

\newtheorem{lemma}{Lemma}[section]
\newtheorem{theorem}{Theorem}[section]


%\def\udcs{517.929} %����� ����� ���������� ��� ����� ������

\def\currentyear{2026}
\def\currentvolume{18}
\def\currentissue{2}
\lastpage{65}
\urle{umj.ufaras.ru/index.php/umj/article/view/2026-18-2-55}
\renewcommand{\baselinestretch}{1.00}
\firstpage{55}


\renewcommand{\leq}{\leqslant}
\renewcommand{\geq}{\geqslant}

\renewcommand{\le}{\leqslant}
\renewcommand{\ge}{\geqslant}

%%%%% ��������� ������� %%%%%

\def\tht{\theta}
\def\Om{\Omega}
\def\om{\omega}
\def\e{\varepsilon}
\def\g{\gamma}
\def\G{\Gamma}
\def\l{\lambda}
\def\p{\partial}
\def\D{\Delta}
\def\bs{\backslash}
\def\k{\varkappa}
\def\E{\mbox{\rm e}}
\def\a{\alpha}
\def\b{\beta}

\def\si{\sigma}
\def\Si{\Sigma}
\def\d{\delta}
\def\L{\Lambda}
\def\z{\zeta}
\def\vs{\varsigma}
\def\r{\rho}
\def\vp{\varphi}
\def\vw{\varpi}
\def\vr{\varrho}
\def\vt{\vartheta}
\def\h{\widehat}
\def\Odr{O}
\def\H{W_2}
\def\Ho{\mathring{W}_2}
\def\Hloc{W_{2,loc}}
\def\Hper{W_{2,per}}
\def\Hoper{\mathring{W}_{2,per}}
\def\Hinf{W_\infty}

\def\mD{{D}}
\def\mR{{R}}


\def\Th{\Theta}
\def\I{\mathrm{I}}
\def\iu{\mathrm{i}}
\def\di{\,d}



\def\cH{\mathcal{H}}
\def\cS{\mathcal{S}}

\def\la{\langle}
\def\ra{\rangle}

\def\cB{\mathcal{B}}
\def\cP{\mathcal{P}}
\def\cI{\mathcal{I}}
\def\cL{\mathcal{L}}
\def\cA{\mathcal{A}}
\def\cM{\mathcal{M}}
\def\cQ{\mathcal{Q}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\DeclareMathOperator{\RE}{Re}
\DeclareMathOperator{\IM}{Im}
\DeclareMathOperator{\spec}{\sigma}
\DeclareMathOperator{\discspec}{\sigma_{disc}}
\DeclareMathOperator{\essspec}{\sigma_{ess}}
\DeclareMathOperator{\dist}{dist}
\DeclareMathOperator{\supp}{supp}
\DeclareMathOperator{\sgn}{sgn}
\DeclareMathOperator{\rank}{rank}
\DeclareMathOperator*{\essinf}{ess\,inf}
\DeclareMathOperator{\rot}{rot}
\DeclareMathOperator{\dvr}{div}
\DeclareMathOperator*{\esssup}{ess\,sup}
\DeclareMathOperator{\col}{col}
\DeclareMathOperator{\vraisup}{vraisup}

\DeclareMathOperator{\Dom}{\mathfrak{D}}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{999}

\bibitem{AMR2002} N.V. Azbelev, V.P. Maksimov, L.F. Rakhmatullina.
 {\it  Introduction to the Theory of Functional Differential Equations. Methods and Applications.} Inst. Comput. Studies, Moscow (2002); \textsl{English translation:}  Hindawi Publishing Corporation, New York (2007).
 
 \noindent
 \hrr{http://downloads.hindawi.com/books/9789775945495.pdf}

\bibitem{BPP2013} A.G. Butkovskii, S.S. Postnov, E.A. Postnova. \textit{
Fractional integro--differential calculus and its control--theoretical applications. I: Mathematical fundamentals and the problem of interpretation}
// Autom. Remote Control \textbf{74}:4, 543--574 (2013). \hrr{https://doi.org/10.1134/S0005117913040012}


\bibitem{ZK1968}
 P.P.   Zabre\u{\i}ko, A.I. Koshelev,  M.A. Krasnosel'ski\u{\i}, S.G. Mikhlin, L.S. Rakovshchik, V.Ya. Stet'senko. {\it
Integral Equations.} Nauka, Moscow (1968); \textsl{English translation:}
Noordhoff International Publishing, Leyden, The Netherlands  (1975).

\bibitem{Kr1971}
S.G.  Krein. {\it Linear Equations in Banach Spaces}. Nauka, Moscow (1971); \textsl{English translation:} Birkh\"auser, Boston (1982).


\bibitem{MR1993}
  V.P.  Maksimov, A.N. Rumyantsev. \textit{
Boundary value problems and problems of pulse control in economic dynamics. Constructive study} // Russ. Math. \textbf{37}:5, 48--62 (1993).


\bibitem{M2003} V.P. Maksimov. {\it Issues of general theory of functional--differential equations. Selected works}. Perm State Univ., Perm (2003). (in Russian).

\bibitem{M2014}  V.P. Maksimov.    {\it Modern mathematical methods in economics. Control problems and boundary value problems for linear systems.}  Perm State National Research University, Perm (2014). (in Russian).

\bibitem{M2019} V.P. Maksimov.  {\it On the construction and estimates of the Cauchy matrix for systems with aftereffect} //  Trudy Inst. Matem. Mekh. UrO RAN  {\bf 25}:3,  153--162  (2019). (in Russian). 
    
    \noindent\hrr{https://doi.org/10.21538/0134-4889-2019-25-3-153-162}

\bibitem{M2024} V.P. Maksimov. {\it The Cauchy matrix of a system with fractional derivative and aftereffect} // Appl. Math. Control Sci.  3, 53--63 (2024). \hrr{https://doi.org/10.15593/2499-9873/2024.3.04 }

\bibitem{PGU2022} K.B. Mansimov, Zh.B. Akhmedova. {\it Analogue of Pontryagin maximum principle in optimal control problem for system of differential equations with Caputo fractional derivative and multi--point cost function} // Vestnik Permskogo Univ. Matem. Mekh. Inform.  3(58), 5--10 (2022).

\bibitem{N2000} A.I. Nakhushev. {\it Elements of fractional calculus and their applications}.    Inst. Appl. Math. Automation of Kabardin--Balkar Scientific Center, RAS, Nalchik (2000). (in Russian).	

\bibitem{TTar2017} V.V. Tarasova, V.E. Tarasov. {\it Risk aversion for investors with memory: Hereditary generalizations of the Arrow~---~Pratt measure
 } // Finansovy Zhurnal 2, 46--63 (2017). (in Russian).

\bibitem{Agarwal2024} R.P. Agarwal, S. Hristova,  D. O'Regan.  {\it Impulsive control of variable fractional--order multi--agent systems} // Fractal and Fractional. {\bf 8}:5, 259 (2024). \hrr{https://doi.org/10.3390/fractalfract8050259}

\bibitem{A1986} A.V. Anokhin. {\it On linear impulse systems for functional-differential equations} // Sov. Math., Dokl. {\bf 33}, 220--223 (1986).

\bibitem{AR1996} N.V. Azbelev, L.F. Rakhmatullina. {\it  Theory of linear abstract functional differential equations and applications} // Mem. Diff. Eq. Math. Phys. {\bf 8}, 1--102 (1996). \hrr{https://emis.de/ft/46166}

\bibitem{Gom2023} M.I. Gomoyunov. {\it  On optimal positional strategies in fractional optimal control problems}  // in Proceedings of ``Mathematical optimization theory and operations research. 22nd international conference, MOTOR 2023,'' Spinger, Cham  255--265 (2023). 
    
    \noindent \hrr{https://doi.org/10.1007/978-3-031-35305-5\_17}

\bibitem{KST2006} A.A. Kilbas, H.H. Srivastava, J.J. Trujillo. {\it Theory and Applications of Fractional Differential Equations.} Elsevier, Amsterdam  (2006).

\bibitem{M2006} V.P. Maksimov. {\it Theory of functional differential equations and some problems in economic dynamics} // in ``Proceedings of the International Conference on Differential and Difference Equations and their Applications'',  757--765 (2006).

\bibitem{M2020} V.P. Maksimov.   {\it On a class of linear continuous-discrete systems with discrete memory} //  Vestn. Udm. Univ. Mat. Mekh. Komp. Nauki  {\bf 30}:3, 385--395  (2020). \hrr{https://doi.org/10.35634/vm200303}

\bibitem{Rah2020} M. Rahaman, S.P. Mondal, A.A. Shaikh, A. Ahmadian, N. Senu, S. Salahshour. {\it Arbitrary--order economic production quantity model with and without deterioration: generalized point of view} // Adv. Difference Equ.  {\bf 2020}, 16 (2020).
\hrr{https://doi.org/10.1186/s13662-019-2465-x}

\bibitem{Reh2018} H.U. Rehman, H. Darus., J. Salah. {\it A note on Caputo's derivative operator interpretation in economy} // J. Appl. Math. {\bf 2018}, 1260240  (2018).
\hrr{https://doi.org/10.1155/2018/1260240}

\bibitem{STV1979}	S. Schwabik, M. Tvrd\'y, O. Vejvoda. {\it Differential and Integral Equations: Boundary Value Problems and Adjoints.} D. Reidel Publishing Company Dordrecht, in co-ed. with Academia, Praha (1979).

\bibitem{Tar2020}  V.E. Tarasov. {\it Non-linear macroeconomic models of growth with memory} // Mathematics {\bf 8}:11, 2078 (2020). \hrr{https://doi.org/10.3390/math8112078}

\bibitem{WFZ2016} J. Wong, M. Fe\v{c}kan, Y. Zhou.  {\it  A survey on impulsive fractional differential equations} // Fract. Calc. Appl. Anal. {\bf 19}:4, 806--831 (2016). \hrr{https://doi.org/10.1515/fca-2016-0044}

\end{thebibliography}


%��� \udcs
 \thispagestyle{empty}

\title[On control problems for fractional systems]
{On control problems for \\ fractional systems with aftereffect}


\author{V.P. Maksimov}
\address{Vladimir Petrovich Maksimov,
\newline\hphantom{iii} Perm State National Research University,
\newline\hphantom{iii} Bukireva str. 15,
\newline\hphantom{iii} 614068, Perm, Russia}
\email{maksimov@econ.psu.ru}

\thanks{\sc V.P. Maksimov, On control problems for fractional systems with aftereffect}
\thanks{\copyright \ Maksimov V.P. \ 2026}
\thanks{\it Submitted February 17, 2025.}





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\maketitle
{
\small
\begin{quote}
\noindent{\bf Abstract.}  We consider dynamical models with aftereffect described by functional--differential equations with fractional derivatives. These models encompass processes, in which the system  state may change abruptly at certain points in time, which is interpreted as the result of impulse effects (shocks). The trajectories of such systems may have discontinuities at certain points in time, and between these points the behavior of system   is described by differentiable functions, which satisfy the equation in the usual sense. We pose a general control problem for a given system. We formulate solvability conditions for this problem   in the class of impulse controls, $L_2$--controls, and their hybrids. The proposed approach to studying systems with fractional derivatives is based on the systematic use of abstract functional--differential equation theory and offers certain advantages for studying systems and processes with aftereffects.

\medskip

\noindent{\bf Keywords:} functional--differential systems with fractional derivatives, systems with aftereffect, impulse systems, control problems.

\medskip


  \noindent{\bf Mathematics Subject Classification:}  34K06, 34K30, 34K35, 34K45
\end{quote}
}






\vskip20pt


\section{Introduction}


Despite the abundance and diversity of results obtained over the past 10--15 years, equations and systems with fractional derivatives continue attracting the attention of  both Russian and international scientists. The total number of papers in this area has exceeded one hundred thousands. Furthermore, the range of applied problems, for which the use of fractional derivatives is not only justified but also expedient, continues to expand.
Along with the traditional areas of application of fractional derivative models mentioned in well--known review \cite{BPP2013}, which are relaxation processes in dielectrics, the behavior of electrochemical media, semiconductor physics, plasma physics, processes in micro-- and nanostructured media, signal processing, new application areas   emerge, in particular, ones  related to dynamic economic models \cite{Tar2020}, \cite{TTar2017}, \cite{Rah2020}, \cite{Reh2018}.



We mention only relatively new works related to control problems. In \cite{Agarwal2024}, there were studied models of multi--agent systems, in which the agents dynamics are described by a variable--order Caputo fractional derivative. There was considered the case, in which information exchange between agents occurs only at an initially specified time. Two cases of multi--agent systems were studied, leaderless and with a leader. The right hand sides of the equations in both cases were described by locally defined operators (Nemytskii operators) without memory. We also mention works devoted to obtaining analogs of Pontryagin maximum principle for optimal control problems in systems with fractional derivatives, see, for example, \cite{PGU2022}.

We   separately mention a series of studies devoted to the development of methods for constructing optimal strategies based on the feedback principle in differential games and optimal control problems, see  \cite{Gom2023} and the references therein.

In contrast to  most studies devoted to impulse control problems (see review \cite{WFZ2016}), we consider systems with fractional derivatives, in which not only the fractional differentiation operator, but also the operator defining the right hand side of the equation possesses memory, which seems more natural. Furthermore, when formulating the control problem, we use a rather general target functional, including cases of multi--point and integral targets, as well as an operator implementing programmatic controls, which also possesses memory.

\vskip20pt

\section{Preliminaries}

Let ${\bf D}$ and  ${\bf B}$ be Banach spaces,  and ${\bf D}$ be isomorphic to the direct product $\bf B \times {\mathds R}^n$  ($ \bf D \simeq   \bf B \times {\mathds R}^n$). Hereinafter we suppose that the norm of an element  $\{z, a \}     \in \bf B \times {\mathds R}^n$ is defined by the identity
$$
\|\{z,a  \} \|_{\bf B \times {\mathds R}^n} =\|z\|_{\bf B}+ | a |_{{\mathds R}^n},
$$
where $| \cdot |_{{\mathds R}^n}$ is a norm in  ${\mathds R}^n$.

The equation
\begin{equation} \label{eq:1}
{\mathcal L}x = f
\end{equation}
with a linear bounded operator  ${\mathcal L}:{\bf D}\to {\bf B}$ is called a linear abstract functional--differential equation (AFDE). The theory of equation~\eqref{eq:1} was systematically presented in  \cite{AR1996}, \cite{AMR2002}. We fix an isomoprhism
 $J=\{\Lambda, Y\}:\bf B \times
\mathds R^n \to \bf D$ and denote the inverse operator  $J^{-1}=[\delta,r]$.
Here  $\Lambda: \bf B \to \bf D$, $Y:{\mathds R}^n \to \bf D$ and
$\delta:\bf D \to \bf B$, $r: \bf D \to {\mathds R}^n$ are the corresponding components of operators
 $J$   $J^{-1}$:
\begin{align*}
&J\{z,a\}= \Lambda z+Y a \in {\bf D},\quad z\in {\bf B},\quad a \in {\mathds R}^n,\\
&J^{-1}x=\{\delta x, rx\} \in {\bf B} \times {\mathds R}^n,\quad x \in {\bf D}.
\end{align*}

For each $\{z,a\} \in \bf B \times \mathds R^n$ the system
$$
\delta x= z,\qquad rx= a
$$
has a unique solution
\begin{equation} \label{eq:2}
x=\Lambda z+Ya.
\end{equation}
This identity implies the representation for the operator $\mathcal L$:
$${\mathcal L}x={\mathcal L}(\Lambda z+Y a)={\mathcal L}\Lambda z+{\mathcal L}Ya=Qz+Aa,$$
where the so--called principal part of operator  ${\mathcal L},$ an operator $Q:\bf B \to \bf B$ and a finite--dimensional operator  $A:{\mathds R}^n \to \bf D$ are defined by the identities  $Q={\mathcal L}\Lambda$ and $A={\mathcal L}Y.$ The general theory of equation  \eqref{eq:1} was constructed under the assumption  $Q$ is a Fredholm operator, which is the sum of an invertible operator and a compact operator \cite{Kr1971}.


The general boundary value problem is the system
\begin{equation} \label{eq:3}
{\mathcal L}x=f,\qquad \ell x =\beta,
\end{equation}
where  $\ell =(\ell^1,\ldots,\ell^N): {\bf D} \to {\mathds R^N}$ is a linear bounded vector functional with linearly independent components.  The boundary value problem \eqref{eq:3} is one of the central objects of the theory of AFDEs.
In the case, when $N=n$ and  problem \eqref{eq:3} is uniquely solvable for each $f \in {\bf B} $ and $\beta \in {\mathds R}^n$, the solution can be represented as
\begin{equation} \label{eq:4}
x=G_{\ell}f+X \beta.
\end{equation}
The operator $G_{\ell}:{\bf B} \to {\bf D}$ is called the Green operator of problem \eqref{eq:3}, and the operator $X:{\mathds R}^n \to {\bf D}$ is called the fundamental vector.
A necessary and sufficient condition for the unique solvability of problem \eqref{eq:3} is the invertibility of the matrix $\ell X$.
In the special case $\ell = r$  problem \eqref{eq:3} is called the principal boundary value problem.


We consider the abstract control problem
\begin{equation} \label{eq:5}
{\mathcal L}x=Fu+f,\qquad rx= a,\qquad \ell x=\beta,
\end{equation}
where the control $u$ belongs to the Hilbert space $\bf H,$ $F:{\bf
H} \to {\bf B}$ is a linear bounded operator, $\ell = (\ell_1,\ldots\ell_N)$ is an a target vector functional that determines the target, $\ell x = \beta.$ We  assume that the principle boundary value problem
\begin{equation} \label{eq:6}
{\mathcal L}x=f,\qquad rx= a
\end{equation}
is uniquely solvable.  To formulate a theorem on unique solvability of problem  \eqref{eq:5},  we define a linear bounded functional $\lambda_i :{\bf H} \to {\mathds R}, i=1,\ldots,N,$ by the identity $\lambda_iu=\ell_iG_rFu,$ where $G_r$ is the Green operator of problem \eqref{eq:6}. Obviously, the value of $\lambda_iu$ can be written in the form $\lambda_iu=\langle \nu_i,u \rangle,$ where the brackets
$\langle\,\cdot\,,\,\cdot\,\rangle$ denote the scalar product in $\bf H$, and $\nu_i$ is the element of the space $\bf H$ generating the functional $\lambda_i:{\bf H} \to {\mathds R}.$


\begin{theorem} [\cite{M2006}]
Control problem \eqref{eq:5} is solvable for all $f \in {\bf B}$ and $a \in {\mathds R}^n$, \linebreak $\beta \in {\mathds R}^N $ if and only if the matrix
$W \buildrel{\it def}\over{=}\{\langle \nu_i,\nu_j\rangle\}_{i,j=1,\ldots,N}$ is invertible. The control $u_0=\sum_{i=1}^N \nu_i c_i$, where
$\col(c_i,\ldots,c_N)=W^{-1}[\beta-\ell G_rf -\ell X a],$ resolves problem \eqref{eq:5}.
\end{theorem}

\vskip20pt

\section{Impulse systems}

A permanent interest to impulsive systems with integer order derivatives arose in the middle of 20th century. Their systematic study is associated with many renowned names; references to fundamental results in this area can be found in  \cite[Ch. 3]{AMR2002}. The trajectories of such systems can have discontinuities at certain moments in time,  and between these times the behavior of
  system  is described by differentiable functions, which satisfy an equation in the usual  sense. The modern theory
of impulsive systems is based on the theory of generalized functions (distributions), the foundations of which were laid by S.L.~Sobolev and L.~Schwartz. Another approach to the study of differential equations with discontinuous solutions is associated with so-called ``generalized ordinary differential equations,'' the study of which was initiated by J.~Kurzweil. Nowadays this
theory  is widely developed. According to admitted approaches, impulsive equations are considered in the class
of functions of bounded variation. In this case, the solution is understood as a function of bounded variation satisfying an integral equation with a Lebesgue~---~Stieltjes or Perron~---~Stieltjes integral. Integral equations in the space of functions of bounded variation  became a subject of special interest and were studied in detail in \cite{STV1979} and subsequent works. We recall that a function of bounded variation can be represented as the sum of an absolutely continuous function, a jump function, and a singular component (a continuous function, the derivative of which vanishes almost everywhere). The solutions of equations with impulsive action, which are considered below, do not contain a singular component and can have discontinuities only at a finitely many  given points.


We follow the approach to impulsive systems proposed in \cite{A1986}, which is based on considering equations in a space being  a finite--dimensional extension of the traditional space of absolutely continuous functions. This approach does not employ a complicated theory of generalized functions, it proved to be rich in content,  and it finds many applications in cases, where
the question of the singular component does not arise, in particular, in certain problems of economic dynamics \cite{AR1996}. From the point of view  of AFDE theory, the considered  impulsive systems   are one of the specific representatives of equations considered on finite--dimensional extensions of traditional spaces. Moreover, the features of such equations is determined not only by the features of these spaces, but also by the specific properties of the isomorphism $ J: {\bf D} \to {\bf B} \times {\mathds R}^n$. For equations with integer derivatives, various examples of such equations can be found in   \cite[Ch. 3]{AMR2002}.


We  first recall the construction of the basic trajectory space for impulsive systems with integer order derivatives.

Let ${\bf L}$ be the space of Lebesgue summable functions $z: [0,T] \to {\mathds R}^{n} $ with the norm $$\| z \|_{\bf L} = \int\limits_0^T\,|z(t)|_{{\mathds R}^{n}}\,dt;$$
${\bf AC}$ be the space of absolutely continuous functions $y: [0,T] \to {\mathds R}^{n} $ with the norm $$ \| y \|_{\bf AC} = \|\dot y \|_{\bf L}+| y(0) |_{{\mathds R}^{n}};$$
${\bf L}_{\infty}$ be the space of measurable and essentially bounded functions $z: [0,T] \to {\mathds R}^{n} $ with the norm $$ \| z \|_{{\bf L}_{\infty}} = \vraisup (|z(t)|_{{\mathds R}^{n}}: t \in [0,T]);$$ ${\bf AC}_{\infty}$ be the space of absolutely continuous functions $y: [0,T] \to {\mathds R}^{n} $ with essentially bounded derivative $\dot y$ and  the norm $$\| y \|_{{\bf AC}_{\infty}} = \|\dot y \|_{{\bf L}_{\infty}}+| y(0) |_{{\mathds R}^{n}}.$$

We fix a discrete set of points $t_k \in (0,T)$, $0<t_1<\ldots<t_m<T.$ We consider the space ${\bf D}={\bf DS}(m)$ of functions $x:[0,T] \to {\mathds R^n}$, which can be  represented as
$$x(t)= \int\limits_0^t z(s)\,ds+x(0)+\sum_{k=1}^m \chi_{[t_k,T]}(t)\Delta x(t_k), $$
where $z \in {\bf L}$, $\Delta x(t_k)=x(t_k)-x(t_k-0),$ $\chi_{[t_k,T]}(t)$ is the characteristic function of the interval $[t_k,T]$.
In this case we have
${\bf D} \simeq {\bf L} \times {\mathds R}^{n+mn}$ with the isomorphism $ J=\{\Lambda, Y\} $,
$$ (\Lambda z)(t)=\int\limits_0^t z(s)\,ds; \qquad (Y a)(t)=a^0+\sum_{k=1}^m \chi_{[t_k,T]}(t) a^k; \qquad a = \col(a^0,\ldots,a^m).$$

At the same time, for $ J^{-1} = \{\delta, r\} $ we have
$$ \delta x= \dot x, \qquad rx=\col(x(0),\Delta x(t_1),\ldots,\Delta x(t_m)). $$
The results of studying control problems for a wide class of impulse systems with an ordinary derivative and a trajectory space ${\bf D}={\bf DS}(m)$
we presented in   \cite{MR1993}, \cite{M2006}.

Let us now turn to the definition of the underlying spaces and isomorphisms that allow us to consider impulse systems from the point of view of  AFDE theory in the case of fractional derivatives. We shall employ  two well--known operators, the Caputo fractional differentiation operator of order
  $\alpha \in (0, 1)$ \cite{N2000}, \cite{KST2006},
$$
({\mathcal{D}}^{\alpha}x)(t) = \frac{1}{\Gamma(1-\alpha)} \int\limits_0^t \frac{\dot{x}(s)}{(t - s )^{\alpha}} \,ds,
$$
and the Riemann~---~Liouville fractional integration operator of the same order \cite{N2000}, \cite{KST2006},
$$
({\mathcal{J}}^{\alpha}z)(t) = (\Lambda z)(t)=\int\limits_0^t \frac{(t-s)^{\alpha - 1}}{\Gamma(\alpha)} z(s)\,ds.
$$
Here ${\Gamma(\cdot)}$ is the Euler Gamma function.


The discrete set of points  $t_k \in (0,T),$ $0<t_1<\ldots<t_m<T$  (``moments of impulse action'') is still supposed to be fixed.


We consider the space  ${\bf D}={\bf DS}_{\infty}(m)$  of functions $x:[0,T] \to {\mathds R^n}$ represented as
$$
x(t)= ({\mathcal{J}}^{\alpha}z)(t)   +x(0)+\sum_{k=1}^m\,\chi_{[t_k,T]}(t)\Delta x(t_k),
$$
where $z \in {\bf L}_{\infty}$, $\Delta x(t_k)=x(t_k)-x(t_k-0),$ $\chi_{[t_k,T]}(t)$ is the characteristic function of  segment  $[t_k,T]$. In this case we have
${\bf D} \simeq {\bf L}_{\infty} \times {\mathds R}^{n+mn}$,
\begin{align*}
&(\Lambda z)(t)= ({\mathcal{J}}^{\alpha}z)(t);\qquad (Y a)(t)=a^0+\sum_{k=1}^m \chi_{[t_k,T]}(t) a^k;\qquad a = \col(a^0,\ldots,a^m),\\
&\delta x= {\mathcal{D}}^{\alpha}x;\qquad rx=\col (x(0),\Delta x(t_1),\ldots,\Delta x(t_m)).
\end{align*}

We consider the impulse system
\begin{equation}  \label{eq:8}
({\mathcal{D}}^{\alpha}x)(t)= (\mathcal{T}x)(t)+f(t), \qquad t\in [0,T],
\end{equation}
where
$\mathcal{T}: {\bf DS}_{\infty}(m)  \to {\bf L}_{\infty}$  is a linear bounded Volterra operator  \cite{AMR2002} such that there exists
 $\rho > 0,$ which ensures the inequality
$$
|(\mathcal{T}y)(t)| \leq \rho \displaystyle{\max_{s \in [0, t]}} |y(s)|,  \qquad t \in [0, T],
$$
for all    $ y \in {\bf AC}_{\infty}^{\alpha}.$ Here $ {\bf AC}_{\infty}^{\alpha}$ is the space of functions $ y : [0,T] \to {\mathds{R}}^n $, which can be represented as
$$y(t) = y(0) + ({\mathcal{J}}^{\alpha}z)(t),$$ where $ z \in {\bf L}_{\infty}$, with the norm
$$ \| y \|_{{\bf AC}_{\infty}^{\alpha}} = \|{\mathcal{D}}^{\alpha}y \|_{{\bf L}_{\infty}}+| y(0) |_{{\mathds R}^{n}}.$$ Hereafter we assume that the norm $| \cdot|$ in ${\mathds{R}}^n$ has the monotonicity property:
for all $a=\col(a_1,\ldots,a_n)$, $b=\col(b_1,\ldots,b_n)$ $ \in {\mathds{R}}^n $ such that $|a_i|\leq |b_i|$, $i = 1,\ldots,n,$ we have $|a| \leq |b|$.

We also suppose that
 the operator
$K = \mathcal{T} J^{\alpha }$ is a regular Volterra integral operator \cite[Ch. 5]{ZK1968}
$$ (K z)(t)= \int\limits_0^t K(t,s) z(s)\,ds. $$
It was shown in \cite{M2024} that the operator $(I - K)^{-1}$ is well--defined and can be represented as a convergent Neumann series; hereinafter $I$ is the identity operator.
As it is known \cite[Thm 2.2]{ZK1968}, in this case $(I - K)^{-1} = I + R $, where $R$ is the resolvent operator, which is also a Volterra integral operator
$$
(R f)(t)= \int\limits_0^t R(t,s)f(s)\,ds
$$
with the resolvent kernel  $R(t,s)$.

As an example of the operator $\mathcal{T}$ obeying the above formulated conditions, the following operator serves
\begin{equation} \label{eq:12}
(\mathcal{T}x)(t)= P(t) x_h(t), \qquad h(t) \leq  t,
\end{equation}
where the columns of $(n\times n)$ matrix $P$ belong to the space  ${\bf L}_{\infty},$
\begin{equation}  \label{eq:13}  x_h(t)  = \left\{
         \begin{aligned}
      &x[h(t)], \quad &&t \in [0,T],\\
     &0,\quad &&t \notin [0, T],\\
      \end{aligned}
      \right.
				   					\end{equation}
and the function	$h : [0,\,T]  \to R $ is measurable.


The principal boundary value problem for system \eqref{eq:8} reads
\begin{equation}  \label{eq:14}
{\mathcal{D}}^{\alpha}x = \mathcal{T}x +f; \qquad x(0) = a^0,\qquad \Delta x(t_1) = a^1,\quad \ldots,\quad \Delta x(t_m) = a^m.
\end{equation}
It was shown in \cite{M2024} that for $a^k = 0$, $k = 0,\ldots,m,$ this problem  is uniquely solvable for each  $f \in {\bf L}_{\infty}$ and its solution can be represented as
\begin{equation} \label{eq:15}
x(t)=(C f)(t)=  \int\limits_0^tC(t,s) f(s)\,ds,
\end{equation}
where $C(t,s)$ is the Cauchy matrix defined by the identity
\begin{equation} \label{eq:16}
C(t,s) = \frac{(t-s)^{\alpha - 1 }}{\Gamma(\alpha)}E + \int\limits_s^t \frac{(t-\tau)^{\alpha - 1 }}{\Gamma(\alpha)} R(\tau, s)\,d\tau;
\end{equation}
hereinafter  $E$ is the unit $(n \times n)$ matrix.

We introduce the notation
$$
{\mathcal A}^0(t) = (\mathcal{T}E )(t),\qquad {\mathcal A}^k(t)  = (\mathcal{T} (E \chi_{[t_k,T]}) )(t),\qquad k=1,\ldots,m.
$$
We note that for an $(n \times n)$ matrix $X$ with columns $X_i$ in ${\bf DS}_{\infty}(m)$, the notation $(\mathcal{T} X)(t)$ means the matrix $((\mathcal{T}X_1)(t),\ldots,(\mathcal{T}X_n)(t))$.

We define matrices   $X^0(t)$, $X^1(t)$, $\ldots$, $X^m(t)$  by the identities
\begin{align} \label{eq:19}
&X^0(t) = E+ \int\limits_0^t C(t,s) {\mathcal A}^0(s)\,ds,\\
&X^k(t)= \chi_{[t_k,T]}(t)E+ \int\limits_0^t C(t,s) {\mathcal A}^k(s)\,ds, \quad k=1,\ldots,m. \nonumber
\end{align}

Let us obtain the representation for solution to problem \eqref{eq:14} for all $a^k$, $k=0,\ldots,m.$

\begin{theorem}
Problem   \eqref{eq:14} is uniquely solvable for all $f \in {\bf L}_{\infty}$ and  $a^k \in {\mathds R}^n$, $k=0,1,\ldots,m,$ and its solution  $x \in {\bf DS}_{\infty}(m)$ admits the representation
\begin{equation} \label{eq:20}
x(t) = (X^0(t), X^1(t), \ldots, X^m(t)) \left(\begin{aligned} a^{0} \\ a^{1} \\ \ldots \\  a^m  \end{aligned}\right) +  \int\limits_0^t C(t,s) f(s)\,ds.
\end{equation}
\end{theorem}

\begin{proof} We need to prove only the representation for solution of homogeneous system~\eqref{eq:8} (the case $f =0$). We consider the problem
\begin{equation} \label{eq:21}
{\mathcal{D}}^{\alpha}x = \mathcal{T}x; \qquad x(0) = a^0,\qquad \Delta x(t_1) = 0, \quad \ldots,\quad \Delta x(t_m) = 0.
\end{equation}
The change $y(t) = x(t) - a^0$ reduces it to
\begin{equation} \label{eq:22}
{\mathcal{D}}^{\alpha} y = \mathcal{T}y + \mathcal{T}a^0; \qquad y(0) = 0,\qquad \Delta y(t_1) = 0, \quad \ldots,\quad \Delta y(t_m) = 0.
\end{equation}
For the solution of this problem we have
\begin{equation} \label{eq:23}
y(t) = \int\limits_0^t C(t,s) (\mathcal{T}a^0)(s)\,ds = \int\limits_0^t C(t,s) (\mathcal{T} E)(s)\,ds \,a^0 = \int\limits_0^t C(t,s) {\mathcal A}^0(s)\,ds \,a^0.
\end{equation}
Returning back  to the original phase variable and denoting by
$x^0$ the solution to problem
\eqref{eq:21}, we obtain
\begin{equation} \label{eq:24}
x^0(t)=a^0  + \int\limits_0^t C(t,s) {\mathcal A}^0(s)\,ds \, a^0 = X^0(t) \, a^0.
\end{equation}
Repeating this arguing for the problem
\begin{equation} \label{eq:25}
{\mathcal{D}}^{\alpha}x = \mathcal{T}x; \quad x(0) = 0,\qquad \Delta x(t_1) = a^1, \qquad \Delta x(t_2) = 0, \quad \ldots,\quad\Delta x(t_m) = 0
\end{equation}
with the change $y(t) = x(t) - \chi_{[t_1,T]}(t) a^1$, we obtain the representation for the solution of problem \eqref{eq:25}
\begin{equation} %\label{eq:24}
x^1(t)=\chi_{[t_1,T]}(t) a^1+\int\limits_0^t C(t,s) {\mathcal A}^1(s)\,ds \, a^1= X^1(t)\, a^1.
\end{equation}
We obtain a similar representation for the solution of problem
\begin{align} \label{eq:26}
&{\mathcal{D}}^{\alpha}x = \mathcal{T}x; \qquad x(0) = 0,\qquad \Delta x(t_1) = 0, \, \ldots,\, \Delta x(t_k) = a^k, \, \ldots,\, \Delta x(t_m) = 0:\\
&x^k(t)=\chi_{[t_k,T]}(t) a^k + \int\limits_0^t C(t,s) {\mathcal A}^k(s)\,ds \, a^k =  X^k(t)\, a^k,\quad k=2,\ldots,m. \nonumber
\end{align}
It remains to note that  $ x(t)= \sum_{k=0}^{m} x^k(t) $. The proof is complete.
\end{proof}

\vskip20pt

\section{Impulse control problem}

We impose control problem for system \eqref{eq:8} treating the jumps of trajectories
 $\Delta x(t_k)$ at the time moments $t_k$, $k=1,\ldots,m,$ as control.

We suppose that the initial state of system is given
\begin{equation}  \label{eq:32}
x(0) = a^{0},
\end{equation}
while the target of control is defined by the identity
\begin{equation}  \label{eq:33}
\ell x \equiv \sum_{j=1}^\mu A_j x(\tau_j)+ \int\limits_0^T B(\tau) x(\tau)\,d{\tau}+\sum_{k=1}^m H_k \Delta x(t_k)= \beta \in {\mathds R}^N,
\end{equation}
where  $\tau_j,$ $j=1,\ldots,\mu,$ are fixed points in   $[0, T],$ generally speaking, these points are not related with the points  $t_k$, $k=1,\ldots,m,$ introduced above; $A_j$ and   $H_k$ are constant  $(N \times n)$ matrices, $B(\cdot)$ is an  $(N \times n)$ matrix with summable entries,  $\beta$  is a given constant of target values.

We rewrite representation   \eqref{eq:20} as
\begin{equation} \label{eq:34}
x(t) =  X^0(t)a^{0} + \sum_{k=1}^m  X^k(t) a^{k} + \int\limits_0^t C(t,s) f(s)\,ds.
\end{equation}
In this representation, the vectors $a^{k} \in {\mathds R}^n$, $k=1,\ldots,m$ are free and are to be determined to attain target  \eqref{eq:33}.


To formulate the solvability conditions for e problem \eqref{eq:8}, \eqref{eq:32}, \eqref{eq:33} we introduce the  notation
\begin{align} \label{eq:35}
& \Xi^k=\sum_{j=1}^{\mu} A_j X^k(\tau_j) + \int\limits_0^T  B(\tau) X^k(\tau)\,d\tau + H^k, \quad k=1,\ldots,m,\\
 \label{eq:36}
&\beta_1= \left(\sum_{j=1}^{\mu} A_j X^k(\tau_j)\right)a^{0} + \int\limits_0^T B(\tau) X^0(\tau)\,d\tau \,a^{0},\\
 \label{eq:37}
&\beta_2= \sum_{j=1}^{\mu} A_j \int\limits_0^{\tau_j} C(\tau_j,s) f(s)\,ds + \int\limits_0^T \int\limits_s^T B(\tau) C(\tau,s)\,d\tau \,f(s)\,ds.
\end{align}

\begin{theorem}
The solvability of  problem \eqref{eq:8}, \eqref{eq:32}, \eqref{eq:33} is equivalent to the solvability of linear system
\begin{equation} \label{eq:38}
\sum_{k=1}^m \Xi^k\,a^{k}=\beta - \beta_1 - \beta_2,
\end{equation}
where $(N \times n)$ matrices $\Xi^k$ and the vectors $\beta_1$, $\beta_2 \in {\mathds R}^N$ are defined by identities \eqref{eq:35}, \eqref{eq:36}, \eqref{eq:37}.
Each solution $ (a^{1}_0,\ldots,a^{m}_0)$, $a^{k}_0 \in {\mathds R}^n $ of this system generates a solution $x_0 \in {\bf DS}_{\infty}(m)$ of   problem \eqref{eq:8}, \eqref{eq:32}, \eqref{eq:33}, and
$\Delta x_0(t_k) = a^{k}_0$.
\end{theorem}


\begin{proof}
By means of   representation \eqref{eq:34} we find the value of the target vector functional $\ell$ \eqref{eq:33} on the right hand side of this identity. In order to do this, we sequentially calculate $\ell$ on each of its three terms, denoting these components by $\ell_1 x$, $\ell_2 x$, $\ell_3 x$:
\begin{align*}
\ell_1 x= &\sum_{j=1}^{\mu} A_j X^0(\tau_j) a^{0} +  \sum_{j=1}^{\mu} \sum_{k=1}^m  A_j X^k(\tau_j)a^{k}+\sum_{j=1}^{\mu}  A_j \int\limits_0^{\tau_j} C(\tau_j,s) f(s)\,ds,\\
\ell_2 x =& \int\limits_0^{T}B(\tau) X^0(\tau)\,d\tau \,a^0 + \int\limits_0^{T}B(\tau)\sum_{k=1}^m  X^k(\tau)d\tau\, a^k\\
&+ \int\limits_0^{T}B(\tau) \int\limits_{0}^{\tau} C(\tau, s)f(s)ds\,d\tau,\\
\ell_3 x=&\sum_{k=1}^{m}  H^k a^k.
\end{align*}
Equating the like terms in the expression  $\ell x = \ell_1 x  + \ell_2 x  + \ell_3 x$ with respect to the factors $a^{0}$,  $a^{k}$ and the integral operators acting on  $f(\cdot)$, interchanging the summation order in the  double sums and the integration order in double integrals, we express target identity
  \eqref{eq:33} in form \eqref{eq:38}. The proof is complete.
\end{proof}

We note that in the case $nm > N$, the freedom in choosing the impulse control variables is excessive. In this case, the target conditions can include additional conditions requiring the jumps of individual trajectory components to be zero ($\Delta x_i(t_k) = 0$).

We do not discuss the practical implementation of impulse effects, i.e., the corresponding trajectory jumps. We note only that in economic dynamics problems, such jumps have a natural interpretation. For example, a negative jump in some trajectory component means an  one--time transfer of production assets (capital) from the corresponding industry to another industry, for which the trajectory component jump at the same time will be positive. If the absolute values of the aforementioned jumps are equal, such a transfer of assets does not require additional resources. For systems with integer derivatives, these issues were discussed in detail in   \cite{MR1993}, \cite[Part 2, Lect. 1]{M2003}, \cite[Ch. 3]{M2014}.

\vskip20pt

\section{Mixed control problem}

We consider the control functional--differential system with fractional derivative
\begin{equation}  \label{eq:41}
{\mathcal{D}}^{\alpha}x = \mathcal{T}x+ Fu + f,
\end{equation}
where $F: {\bf L}_2  \to  {\bf L}_{\infty}$ is a linear bounded Volterra operator implementing control actions $u \in {\bf L}_2$. Here
${\bf L}_2$ is the space of square--summable functions $v: [0,T] \to {\mathds R}^q$ with the scalar product
$ \left\langle u,v \right\rangle = \int\limits_0^T u^{\prime}(t) v(t)\,dt,$ and $(\cdot)^{\prime}$ denotes the transposition.

We still continue to suppose  that the initial state of the system is given by \eqref{eq:32}
and the control target is given by  \eqref{eq:33}.





In the case under consideration, the entire set of trajectories of the system \eqref{eq:41} is described by the identity
\begin{equation} \label{eq:42}
x(t)= X^0(t) a^{0} + \sum_{k=1}^m X^k(t)a^{k} + \int\limits_0^t C(t,s) (Fu)(s)\,ds  +  \int\limits_0^t C(t,s) f(s)\,ds.
\end{equation}
In this notation, the mixed control action contains an impulse finite--dimensional component $ (a^{1},\ldots,a^{m})^{\prime} \in {\mathds R}^{nm}$ and a component $ u \in {\bf L}_2$.

To formulate the solvability conditions for  control problem \eqref{eq:41},\eqref{eq:32},\eqref{eq:33}, we introduce an additional notation.
By $\Theta_0$ we denote the $(N \times q)$ matrix, which defines  the linear bounded vector function $\Theta_0 : {\bf L}_2 \to {\mathds R}^{N}$:
\begin{equation}
\Theta_0 u =  \int\limits_0^T \Theta_0 (s) u(s)\,ds = \int\limits_0^T \int\limits_s^T B(\tau) C(\tau,s)\,d\tau (Fu)(s)\,ds.
\end{equation}
Let $\Theta_j$ --- $(N \times q)$  be a matrix, which defines a linear bounded vector functional $\Theta_j : {\bf L}_2 \to {\mathds R}^{N}$:
\begin{equation}
\Theta_j u = \int\limits_0^T \Theta_j (s) u(s)\,ds = \int\limits_0^T \chi_j (s) C(\tau_j,s) (Fu)(s)\,ds,
\end{equation}
where $\chi_j (\cdot)$  is the characteristic function of segment $[0, \tau_j]$, $j=1,\ldots,\mu,$
\begin{equation}
M(s)= \Theta_0 (s) + \sum_{j=1}^{\mu}  A_j \Theta_j (s);\qquad W= \int\limits_0^T M(s) M^{\prime}(s)\,ds.
\end{equation}

\begin{theorem}
Control problem \eqref{eq:41},\eqref{eq:32},\eqref{eq:33}
is solvable if and only if the linear algebraic system
\begin{equation} \label{eq:45}
\sum_{k=1}^m  \Xi^k a^k + W d = \beta - \beta_1 - \beta_2,
\end{equation}
where $\Xi^k$, $k=1,\ldots,m$  are   $ (N \times n)$ matrices, and the vectors $ \beta_1$, $\beta_2 \in {\mathds R}^N$
are defined by identities \eqref{eq:35}, \eqref{eq:36}, \eqref{eq:37}, respectively,
is solvable with respect to the $(nm+N)$--dimensional vector $(a^1,\ldots,a^m,d)^{\prime}$.
Each solution $(a_0^1,\ldots,a_0^m,d_0)^{\prime}$ of system \eqref{eq:45}
defines a mixed control that solves the control problem \eqref{eq:41}, \eqref{eq:32}, \eqref{eq:33}:
$\Delta x(t_k) =a_0^k$, $k=1,\ldots,m,$  $u(t)= M^{\prime}(t) d_0$.
\end{theorem}

\begin{proof} The introduction of $u \in {\bf L}_2$ into the control system leads to an additional term $\int\limits_0^t C(t,s) (Fu)(s)\,ds $
in  description \eqref{eq:42} of the set of all trajectories and the need to calculate the value of the target vector functional on this term. Such a  calculation gives two additional terms relative to the case of impulse control (Theorem 4.1):
\begin{align} \label{eq:46}
&\ell_4 x= \int\limits_0^T B(\tau)\int\limits_0^{\tau} C(\tau,s) (Fu)(s)\,ds d{\tau},\\
 \label{eq:47}
&\ell_5 x= \sum_{j=1}^{\mu}  A_j \int\limits_0^{\tau_j}  C(\tau_j,s) (Fu)(s)\,ds .
\end{align}
In view of   the introduced notations $\Theta_0 $ and $\Theta_j $ we obtain
\begin{equation} \label{eq:48}
\ell_4 x+\ell_5 x=\int\limits_0^T [ \Theta_0(s)+\sum_{j=1}^{\mu}  A_j \Theta_j(s) ]  u(s)\,ds =\int\limits_0^T M(s) u(s)\,ds .
\end{equation}
Let $ L $ be the linear manifold spanned by the columns of   matrix $ M^{\prime} $, and $ { L}^{\bot} $ be its orthogonal complement in $ {\bf L}_2 $. We take an arbitrary linear combination of the columns of $ M^{\prime} $: $ g = M^{\prime} d $, $ d \in {\mathds R}^N $. Clearly, $ g \in { L} $. By the theorem on orthogonal decomposition in Hilbert space, each control $ u $ can be represented as $ u = g + h $, where $ h \in { L}^{\bot} $. Moreover, $ \int \limits_0^T M(s) h(s) \, ds = 0, $
and the term $ h $ makes zero contribution to the value of   target vector functional. Thus, the term $\int\limits_0^T M(s) u(s)\,ds $ in the values of the target vector functional can be replaced by the term $W d$ without loss of generality, and we obtain the system \eqref{eq:45}. The proof is complete.
\end{proof}


We note that in the case of invertible matrix  $W$ the control
$$
u(t) = M^{\prime}(t)  W^{-1} (\beta  -  \beta_1  -  \beta_2  - \sum_{k=1}^m  \Xi^k a^k)
$$
solves problem  \eqref{eq:41}, \eqref{eq:32}, \eqref{eq:33} for all $a^k  = \Delta x(t_k)$,  $k=1,\ldots,m.$


Reduction of control problem \eqref{eq:41}, \eqref{eq:32}, \eqref{eq:33} to   linear algebraic system \eqref{eq:45} gives an opportunity to apply constructive  methods developed for control problems of functional--differential systems with integer order derivatives (see \cite{MR1993}; \cite[Ch. 6]{AMR2002}), and implemented by using modern computer technology,  reliable computer experiment  (RCE). Note that the key point of this reduction is the usage of a representation of the set of all trajectories of the control system using the Cauchy matrix. In the general case, the absence of an explicit representation of the Cauchy matrix leads to the need to use its approximations with a guaranteed error estimate within the framework of RCE. Methods and algorithms for such construction for some very broad classes of functional--differential systems were described in \cite{M2019}, \cite{M2020}.



\vskip30pt




\end{document}


